Author's response to reviews

Title: Magnetic Resonance Myocardial Perfusion Imaging in diagnosis of functionally significant obstructive coronary artery disease. A systematic review protocol

Authors:

Rebabonye B Pharithi Dr (rpharithi@gmail.com)
Meela Martin Dr (mathinase@gmail.com)
Michael Newell Dr (michael.newell@nuigalway.ie)

Version: 2 Date: 4 November 2013

Author's response to reviews:

Dr R. Pharithi have drafted the manuscript protocol, and written it up. Dr Martinus Meela have assisted in the drafting throughout, and writing of the final paper. Dr Michael Newell was the supervisor and had a lot of input in the search strategy and protocol design.